Author's response to reviews

Title: Addressing maternal healthcare through demand side financial incentives: experience of Janani Suraksha Yojana Program in India

Authors:

Saji S Gopalan (sajisaraswathyg@gmail.com)
Varatharajan D (sajisaraswathyg@yahoo.com)

Version: 2 Date: 10 January 2012

Author's response to reviews:

Dear Editor,

Thanks a lot for the initial consideration of the manuscript. We have incorporated the suggestion required by the journal on "ethical approval" at the end of the Methods section on page 8, 1st paragraph.

Sincerely,

S S Gopalan (on behalf of the authors)